Final Report for Project Name

Student 1^1, Student 2, Student 3, Student 4 Department of Computer Science Michigan State University

Customer: CustomerName, Company

Software Engineering Capstone, CPS479 Instructor: Dr. B. Cheng Teaching Assistant: E. Y. Wang

Spring 1996

1Project Manager 1

Contents

1 Project Description 1 1.1 Motivation . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 1 1.2 Overview of System Design . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 1

2 Techniques Learned and Applied 2 2.1 Methods and Processes . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 2 2.2 Tools . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 2

3 Conclusions and Future Investigations 3 3.1 Functionality Limitations . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 3 3.1.1 System Limitations . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 3 3.1.2 Properties Facilitating System Use . . . . . . . . . . . . . . . . . . . . . . . . 3 3.2 Reflections of Project . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 3 3.3 Future Investigations . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 3

A Changes and Addendums to Deliverables 4 A.1 Requirements Analysis . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 4 A.2 System Design . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 4 A.3 Object Design . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 4

B User Manual 5 1 Introduction . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 5 1.1 Hardware Requirements . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 5 1.2 Software Requirements . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 5 2 Functionality Description . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 5 3 User Scenarios . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 5 4 Installation Procedure . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 5 5 References . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . . 5

Chapter 1

Project Description

1.1 Motivation 1.2 Overview of System Design

Chapter 2

Techniques Learned and Applied

2.1 Methods and Processes 2.2 Tools

Chapter 3

Conclusions and Future Investigations

3.1 Functionality Limitations

3.1.1 System Limitations 3.1.2 Properties Facilitating System Use

3.2 Reflections of Project 3.3 Future Investigations

Appendix A

Changes and Addendums to Deliverables

A.1 Requirements Analysis A.2 System Design A.3 Object Design

Appendix B

User Manual

1 Introduction

1.1 Hardware Requirements 1.2 Software Requirements

2 Functionality Description 3 User Scenarios 4 Installation Procedure 5 References